

# This item is the archived peer-reviewed author-version of:

The evolution of debt policies : new evidence from business startups

# **Reference:**

Hanssens Juergen, Deloof Marc, Vanacker Tom.- The evolution of debt policies : new evidence from business startups Journal of banking and finance - ISSN 0378-4266 - 65(2016), p. 120-133 Full text (Publisher's DOI): https://doi.org/10.1016/J.JBANKFIN.2016.01.008 To cite this reference: https://hdl.handle.net/10067/1318730151162165141

uantwerpen.be

Institutional repository IRUA

# The Evolution of Debt Policies: New Evidence from Business Startups

By Jürgen Hanssens\*, Marc Deloof\*\* and Tom Vanacker\*\*\*

We thank two anonymous *JBF* reviewers, Paolo Colla, Sigitas Karpavicius, Bart Lambrecht, Klaas Mulier, William Megginson, Sofie De Prijcker and Armin Schwienbacher for constructive feedback. The paper further benefited from presentations at the 2014 Belgian Entrepreneurship Research Day (Ghent), the 2014 Belgian Financial Research Forum (Louvain-la-Neuve), the 2014 Corporate Finance Day (Paris), the 2014 RENT Conference (Luxembourg), the 2015 Economics of Entrepreneurship and Innovation Workshop (Trier) and the 2015 Financial Management Association European Meeting (Venice). We acknowledge the financial support of Ghent University's Special Research Fund, the National Bank of Belgium and the Hercules Fund (AUGE/11/13).

\* Corresponding author. Ghent University, Department of Accounting, Corporate Finance and Taxation, Sint-Pietersplein 7, 9000 Gent, Belgium, phone: +32-9-264 35 45, e-mail: Jurgen.Hanssens@UGent.be.

<sup>\*\*</sup> University of Antwerp and Antwerp Management School, Department of Accounting and Finance, Prinsstraat 1, 2000 Antwerp, Belgium, phone: +32-3-264 41 69, e-mail: marc.deloof@uantwerpen.be.

<sup>\*\*\*</sup> Ghent University, Department of Accounting, Corporate Finance and Taxation, Sint-Pietersplein 7, 9000 Gent, Belgium, phone: +32-9-264 79 60, e-mail: TomR.Vanacker@UGent.be.

# The Evolution of Debt Policies:

# **New Evidence from Business Startups**

# **ABSTRACT**

We investigate the evolution of entrepreneurial firms' debt policies over a period of 15 years after startup, considering leverage, debt specialization, debt maturity and debt granularity. Our analysis is based on a unique sample covering all non-financial Belgian firms founded between 1996 and 1998. We find that the debt policy of entrepreneurial firms is remarkably stable over time. The debt policy in the initial year of operation is a very important determinant of future debt policies, even after controlling for traditional contemporaneous determinants. The founder-CEO has an important impact on the stability of debt policies: the influence of initial debt policies on future debt policies is significantly reduced when the founder-CEO is replaced or when (s)he dies. Combined, our findings support imprinting theory.

# JEL classification: G32

Keywords: capital structure, debt structure, entrepreneurial firms, founder-CEOs

### **1. Introduction**

A number of studies have found that over time, the leverage ratio of listed firms (Lemmon, Roberts and Zender, 2008; Welch, 2004; Wu and Yeung, 2012) and established private firms (Hanousek and Shamshur, 2011) contains an important stable component. If past leverage ratios have possible bearing on future leverage ratios, a logical place to start a study of the evolution of leverage is the earliest phase of a firm's existence, i.e., its founding. However, while startups rely on debt financing to a greater extent than often recognized (Cassar, 2004; Cumming, 2005; Robb and Robinson, 2014), no study has yet examined the evolution of leverage in early-stage firms. Moreover, we lack evidence as to whether findings on the dynamics of leverage have implications for a broader range of debt policies, including debt specialization (Colla, Ippolito and Li, 2013), debt maturity (Scherr and Hulburt, 2001) and debt granularity (Choi, Hackbarth and Zechner, 2014). In sum, an investigation of the evolution of entrepreneurial firms' debt policies going back to startup is timely.

How the debt policies of entrepreneurial firms evolve over time remains ambiguous from a theoretical perspective. On the one hand, information-based theories on the evolution of entrepreneurial financing predict that debt policies will change as firms age because firms reveal more information to the market and establish relationships with private debt providers (Berger and Udell, 1995; Petersen and Rajan, 1994, 2002). For instance, Berger and Udell (1998) state that "different capital structures are optimal" (p. 613) and different "sources of finance become important at different points in the financial growth cycle" (p. 622). This view thus suggests that firms' debt policies at startup may have little bearing on their future debt policies. On the other hand, imprinting theory (Boeker, 1989; Stinchcombe, 1965)—which had its roots in the management literature but is also used in economics and finance research (Bertrand and Schoar, 2003; Rotemberg and Saloner, 2000)—suggests that (a) conditions at the time of founding define

initial policies and create internal consensus around the initial policies of the firm, and (b) conditions subsequent to founding tend to preserve previously adopted policies. Imprinting theory thus suggests that firms' debt policies at startup have significant bearing on their future debt policies.

Consistent with imprinting theory, corporate finance research shows how CEOs "imprint their mark" on firms' financial policies, regardless of whether it is optimal (Bertrand and Schoar, 2003, p. 1175). Schoar and Zuo (2014), for instance, show how CEOs with recession experience display more conservative styles in their future career, including holding lower leverage ratios. We therefore consider the influence of founder-CEOs on the evolution of entrepreneurial firms' debt policies. We expect that firms' initial debt policies will exert less influence on future debt policies after the departure of the founder-CEO because entrepreneurial firms may break out of their initial path when new CEOs are appointed. Alternatively, founder-CEO departures may be a consequence of the need for financial reorganization. Using unique data on founder-CEO deaths—exogenous CEO departures unrelated to the need for financial reorganization (or any other unmeasured variable)—we can tease out these alternative explanations.

Scholars have been severely constrained in their efforts to study the evolution of entrepreneurial firms' debt policies from founding because the data required for such an investigation are generally not available (Robb and Robinson, 2014). Belgium, however, represents a unique "laboratory" to study the evolution of firms' debt policies because *all* non-financial firms, including startups, have a legal obligation to annually file detailed financial accounts with the Belgian National Bank. Consequently, we are able to construct a unique database from the population of non-financial firms founded between 1996 and 1998, for which we have detailed financial information for as long as 15 years after startup (i.e., until 2013). Moreover, firms are required to provide detailed information concerning their founding, capital

increases, appointments and resignations and the like in the Belgian Law Gazette, and this information is externally validated by a notary. The Belgian Law Gazette provides unique information about the departure of founder-CEOs in early-stage entrepreneurial firms.

We find that leverage, debt specialization, debt maturity and debt granularity policies in the initial year of operation are statistically and economically significant determinants of future debt policies—even after controlling for traditional contemporaneous determinants. Moreover, variance decomposition analyses show that the variation captured by models that include traditional capital and debt structure determinants is substantially lower than the variation captured by models that only include firm fixed effects. This finding implies that time-invariant and unobservable firm-specific factors present at startup drive the debt policies of entrepreneurial firms to a large extent. We highlight one factor: the founder-CEO. We find that the influence of initial debt policies on entrepreneurial firms' future debt policies significantly declines after the departure of founder-CEOs. To address potential endogeneity of new CEO appointments, we investigate how the death of the founder-CEO affects the evolution of entrepreneurial firms on their future debt policies significantly declines after the death of the policies significantly declines of entrepreneurial firms on their future debt policies significantly declines after the death of the founder-CEO affects the evolution of entrepreneurial firms on their future debt policies significantly declines after the death of the founder-CEO affects the evolution of entrepreneurial firms on their future debt policies significantly declines after the death of the founder-CEO affects the evolution of entrepreneurial firms on their future debt policies significantly declines after the death of founder-CEOs.

Our study contributes to the finance literature in several ways. First, extant research focuses on cross-sectional heterogeneity in the capital structure of entrepreneurial firms by relying on cross-sectional survey data (Cassar, 2004; Cosh, Cumming and Hughes, 2009) or on comparatively short time series of financial data (Robb and Robinson, 2014). We provide unique evidence on the evolution of entrepreneurial firms' debt policies in the 15 years after founding. Second, while an increasing body of research shows the importance of debt financing for new entrepreneurial firms (e.g., Robb and Robinson, 2014), research has only skimmed the surface in terms of exploring the ways new entrepreneurial firms rely on debt financing (Robinson, 2012). We provide first-time evidence on debt specialization, debt maturity and debt granularity choices and their dynamics in very early stage firms. Third, we also contribute to the literature by investigating the effect of founder-CEOs on firm policies (e.g., Bertrand and Schoar, 2003). While several studies have examined the impact of a CEO and of CEO departures on firm policies, especially in large public firms (e.g., Malmendier, Tate and Yan, 2011; Fee, Hadlock and Pierce, 2013), to the best of our knowledge, we are the first to examine the impact of founder-CEO departures (and deaths) on the evolution of entrepreneurial firms' debt policies.

Finally, our study has important ramifications for capital structure theory. New entrepreneurial firms are arguably the most informationally opaque firms (Berger and Udell, 1998). Consequently, we would expect the pecking order theory to be especially relevant in our context because this theory states that the existence of information asymmetry leads to a financing hierarchy. However, the stable component of capital structure cannot be explained by the pecking order theory (Dennis, 2012). Moreover, the static trade-off theory is also unable to explain the stable component of the debt policies because this theory predicts that the financial structure will be rebalanced when it deviates too much from its target (Lambrecht and Myers, 2014). While scholars have used dynamic models to explain the stable component of financial policies in mature public firms by incorporating manager-shareholder agency conflicts (Lambrecht and Myers, 2014; Morellec, Nikolov and Schürhoff, 2012), such models are less suitable for new entrepreneurial firms, in which principal and agents are likely to be the same individuals (Fama and Jensen, 1983). However, the observed stable component of debt policies in entrepreneurial firms is in line with imprinting theory, which argues that important predictors of firms' current financing policies are their financing policies at founding.

The remainder of the paper is organized as follows. Section 2 discusses the research setting. Section 3 describes the data. Section 4 presents the empirical results. Section 5 discusses possible alternative explanations for the findings as well as several extended analyses on subsamples. Section 6 concludes.

#### 2. Research setting

Belgium is a typical example of a Continental European, bank-based financial system in which banks play a central role in mobilizing savings and allocating capital (Demirgüç-Kunt and Levine, 1999). While the Belgian banking sector is well developed, public equity and debt markets play only a minor role in corporate financing. As in other Continental European countries, few firms are quoted on a stock exchange and initial public offerings are rare events. Public debt markets are only accessible for large and mature firms, which are not the focus of this study. The venture capital and private equity market is quite developed in Belgium, compared to other Continental European venture capital and private equity markets (Groh, Liechtenstein and Lieser, 2010)—although less developed than the U.S. and U.K. markets.

During the timeframe of our paper, several important events occurred that may have had a significant impact on the financing of Belgian firms. First, in the period 1997-2003, Belgium experienced a significant wave of bank mergers (e.g., Degryse, Masschelein and Mitchell, 2011), resulting in a heavily concentrated credit market in which four banks provide nearly 80% of total outstanding credit. Second, in 2005 the Belgian government introduced a new tax measure (which was effective from 2006) to reduce the tax advantage of debt financing (e.g., Panier, Pérez-González and Villanueva, 2013). The "notional interest deduction" allows firms subject to Belgian corporate taxes to deduct from their taxable income an amount equal to the interest they would have paid on their "corrected" equity capital if that capital were to be viewed as long-term debt financing. Third, the financial crisis had a negative impact on the Belgian banks. After the collapse of Lehman Brothers in 2008, Fortis Bank—the largest Belgian bank—had to be bailed

out by the Belgian, Luxembourg, and Dutch governments. Subsequently, the other three major Belgian banks had to be rescued by the government. A survey conducted by the Belgian National Bank shows that this led to a net tightening in credit volume, general credit conditions, costs and required collateral for firms.<sup>1</sup>

The occurrence of these events over the timeframe of our study should bias our results against finding stable debt policies. However, despite this bias, we find evidence that the debt policies of entrepreneurial firms in their initial year of operation are important determinants of future debt policies. Before presenting our results in detail, we first discuss our data.

# 3. Method

# 3.1. Sample

The data for this paper are from the Bel-first database. The Bel-first database is compiled by Bureau van Dijk (BvD), one of Europe's leading electronic publishers of business information. Reporting requirements imposed by the Belgian government require all non-financial firms irrespective of their size and age—to annually file detailed financial accounts in a predefined format with the Belgian National Bank.<sup>2</sup> When the financial accounts are filed with the Belgian National Bank, they are processed and checked and subsequently made available to the public. BvD collects these data to compile the Bel-first database. Typically, one annual release of Bel-first covers at most the preceding ten accounting years of each firm. BvD removes firms after at least five years of no reporting data. Therefore, to eliminate this potential survivorship bias, we

<sup>&</sup>lt;sup>1</sup> More information on the survey is available at: http://www.nbb.be/DOC/DQ/kredObs/fr/data/KO\_tarifs.htm.

<sup>&</sup>lt;sup>2</sup> Belgian SMEs are allowed to report abbreviated financial statements when they comply with the following requirements. A firm should (1) employ less than 100 employees on average per year registered or (2) not meet two or more of the following criteria: (i) annual turnover > 6,250,000 euro, (ii) balance sheet total > 3,125,000 euro and (iii) average number of employees > 50. One major difference between abbreviated and complete financial statements is that revenues only have to be disclosed in complete financial statements. However, even the abbreviated statements provide 25 pages of financial information.

compile the database by collecting accounting information from each annual release retrospectively so that we can have the complete history of data for all firms in our sample across the entire sample period.

Firms had to fulfill several criteria to be part of our sample. First, we include all limited liability firms that were legally incorporated in 1996, 1997 or 1998. We select firms founded in multiple years to avoid that idiosyncratic events in a specific founding year would drive our results.<sup>3</sup> To do so, we start from the oldest Bel-first release that is available (the February 1998 release). We use subsequent Bel-first releases to collect data for these firms until the year 2013. Second, firms should have at least 1 employee and less than 50 employees, measured in full-time equivalents, in the year of startup. We use this selection criterion to exclude "ghost" firms (i.e., firms that only exist on paper, primarily for fiscal reasons) and firms that are unlikely to be de novo startups. Third, we only include independent startups because firms that belong to a group structure may have limited discretion over their debt policies. Firms could not be controlled by an external shareholder with an equity stake of 50% or more (except for equity stakes of families, employees and directors) and could not have participations in other firms (ownership > 10%) at startup. Fourth, we exclude financial and government-owned firms because the financing of these firms may be influenced by regulatory issues. Fifth, we only select firm-year observations for which all information needed to calculate our variables is available. It is important to note that our sample not only includes firms that are active in all sample years but also firms that leave the sample over the sample period either due to bankruptcy, acquisitions or buy-outs.

The final sample contains 49,418 firm-year observations, which represent 4,962 firms. Of these 4,962 firms, 2,347 firms are active during all sample years, while 2,615 firms leave the sample either due to bankruptcy, acquisitions or buy-outs.

<sup>&</sup>lt;sup>3</sup> Results on each individual founding year are qualitatively similar to those reported below for the full sample.

The financial data from the Bel-first database are supplemented with information about the departure of founder-CEOs and founder-CEO deaths. This information was, for each of the 4,962 firms, manually collected using the Belgian Law Gazette. In the Belgian Law Gazette, Belgian firms are required to provide detailed information concerning their founding, capital increases, appointments and resignations and the like, and this official information is externally validated by a notary. Of the 4,962 firms, there are 1,907 firms in which the founder-CEO leaves the firm in the first 15 years after startup, and there are 19 firms in which a new CEO is appointed after the death of the founder-CEO.

# 3.2. Variables

We focus on four dependent variables, capturing distinct aspects of firms' debt policies. First, we examine the extent to which a firm's capital structure consists of debt financing by using a firm's leverage ratio. *Leverage* is measured as the ratio of total debt on total assets (e.g., Rajan and Zingales, 1995).<sup>4</sup> Second, we explore the extent to which the debt financing in a firm's capital structure belongs to one type of debt or to a more diversified range of debt sources. Therefore, *debt specialization* is computed using a normalized Herfindahl-Hirschman Index (HHI) of debt usage following a similar procedure as described by Colla et al. (2013). Specifically, we first calculate:

$$SS_{i,t} = \left(\frac{ID_{i,t}}{D_{i,t}}\right)^2 + \left(\frac{BD_{i,t}}{D_{i,t}}\right)^2 + \left(\frac{NBD_{i,t}}{D_{i,t}}\right)^2 + \left(\frac{TD_{i,t}}{D_{i,t}}\right)^2 + \left(\frac{OOD_{i,t}}{D_{i,t}}\right)^2 \quad (1)$$

with  $SS_{i,t}$  the sum of the squared debt type ratios for firm i in year t; *ID*, *BD*, *NBD*, *TD* and *OOD* refer to insider debt, bank debt, non-bank debt (including debt related to payroll or social

<sup>&</sup>lt;sup>4</sup> We also use the ratio of bank debt to total assets, and the results remain qualitatively similar. Rajan and Zingales (1995) discuss the advantages and disadvantages of different leverage measures.

security, taxes), trade debt and other operational debt, respectively; *D* refers to total debt. The Herfindahl-Hirschman Index (HHI) of debt usage is subsequently computed as:

$$HH_{i,t} = \frac{SS_{i,t} - 1/5}{1 - 1/5} \qquad (2)$$

HHI equals one when a firm exclusively uses one type of debt, while HHI equals zero when a firm simultaneously uses all five types of debt in equal proportion. The higher the HHI, the higher is the degree of debt specialization.<sup>5</sup> Third, we examine the maturity structure of firm debt. *Debt maturity* is measured as the percentage of total debt that matures in more than five years (Custodio, Ferreira and Laureano, 2013).<sup>6</sup> Finally, *debt granularity* is used to measure the extent to which a firm spreads out its debt maturity dates (Choi et al., 2014) and is computed following a similar procedure as for debt specialization. Specifically, we first calculate:

$$SS_{i,t} = \left(\frac{D < 1_{i,t}}{D_{i,t}}\right)^2 + \left(\frac{D1 - 5_{i,t}}{D_{i,t}}\right)^2 + \left(\frac{D > 5_{i,t}}{D_{i,t}}\right)^2 \quad (3)$$

with  $SS_{i,t}$  the sum of the squared debt maturity ratios for firm i in year t; D<1, D1-5 and D>5 refer to the amount of debt that matures in one year, the amount of debt that matures between one and five years and the amount of debt that matures in more than five years, respectively; D refers to total debt. The Herfindahl-Hirschman Index (HHI) of debt maturity is subsequently computed as:

$$HH_{i,t} = \frac{SS_{i,t} - 1/3}{1 - 1/3} \quad (4)$$

<sup>&</sup>lt;sup>5</sup> We also use a dummy variable, which equals one if a firm obtains at least 90% of its debt from one debt type and zero otherwise, as an alternative measure of debt specialization (e.g., Colla et al., 2013). Results remain qualitatively similar when using this alternative debt specialization measure.

<sup>&</sup>lt;sup>6</sup> Due to data availability, we could only make a distinction between debt that matures in more than five years, debt that matures between one and five years and debt that matures in one year. In line with Scherr and Hulburt (2001), we therefore also measure debt maturity as the percentage of debt that matures in more than one year. Results remain qualitatively similar when using this alternative debt maturity measure.

HHI is equal to one when a firm does not spread out its debt maturity dates, while HHI equals zero when a firm completely spreads out its debt maturity dates. We multiply this measure with - 1 so that higher values of debt granularity indicate that firms increasingly spread out their debt maturity dates.<sup>7</sup>

The initial values (i.e., the values at startup) of these four distinct aspects of firms' debt policies are our key independent variables. We further construct a dummy variable *founder-CEO departure* and its interaction with initial debt policies to capture the influence of the departure of founder-CEOs on the relationship between entrepreneurial firms' initial debt policies and their subsequent debt policies. Founder-CEO departure equals 1 from the year of the founder-CEO departure onward and 0 otherwise.<sup>8</sup> We also use a dummy variable *founder-CEO departures* on the relationship between entrepreneurial firms' initial debt policies and its interaction with initial debt policies to examine the impact of exogenous founder-CEO departures on the relationship between entrepreneurial firms' initial debt policies and their subsequent debt policies. Founder-CEO death equals 1 from the year of the founder-CEO departures on the relationship between entrepreneurial firms' initial debt policies and their subsequent debt policies. Founder-CEO death equals 1 from the year of the founder-CEO departures on the relationship between entrepreneurial firms' initial debt policies and their subsequent debt policies. Founder-CEO death equals 1 from the year of the founder-CEO death onward and 0 otherwise.

Several control variables that are consistently shown in prior research to be important capital structure and debt structure determinants are included in our analyses, including firm size, profitability, tangibility and growth opportunities (e.g., Titman and Wessels, 1988; Rajan and Zingales, 1995; Brav, 2009; Colla et al., 2013; Choi et al., 2014). *Firm size* is measured as the natural logarithm of total assets. *Profitability* is measured as the amount of earnings before interest and taxes (EBIT) to total assets. *Tangibility* is measured as the ratio of net property, plant

<sup>&</sup>lt;sup>7</sup> In contrast to Choi et al. (2014), we do not take the inverse of the HHI but rather multiply the HHI with -1 because we otherwise lose data (i.e., those firms that completely spread out their debt maturity dates).

<sup>&</sup>lt;sup>8</sup> The available information does not allow us to consider whether the founder-CEO remains a shareholder after (s)he ceases to be the CEO. The fact that we cannot take this into account provides a bias *against* finding any significant effect of founder-CEO departure, as the founder can still exert influence on the firms' policies if (s)he remains a shareholder.

and equipment to total assets. *Firm growth* is measured as the relative growth in total assets (i.e., total assets of the firm in year t minus total assets in year t-1, and this is divided by total assets in year t-1). We further include the *capital expenditures* of firms by measuring the amount of new investments in fixed assets on total assets as an additional proxy for firm growth opportunities (e.g., Brav, 2009).

The creditworthiness of firms is often proxied by ratings given by agencies such as Standard & Poor's, Moody's and Fitch Ratings. The firms in our sample, however, lack such ratings. Therefore, we use the (unlevered) FiTo score, which is a default risk indicator from Graydon. Graydon is the market leader in commercial and marketing information as well as credit and debt management in Belgium. The FiTo score takes values between 0 (financially distressed firms) and 1 (financially healthy firms). The firm-year observations are divided into three categories according to their FiTo score by using dummy variables. *Dummy low creditworthiness* is a dummy variable that is 1 for the bottom 25% of firm-year observations and zero otherwise (low creditworthiness). *Dummy medium creditworthiness* is a dummy variable that is 1 for the firm-year observations and zero otherwise (medium creditworthiness). Finally, *dummy high creditworthiness* is 1 for firm-year observations with a FiTo score above the 75% percentile and zero otherwise (high creditworthiness).

Where appropriate we include year and industry fixed effects in the regressions. We also control for *Industry median leverage (debt specialization, debt maturity* or *debt granularity*), which is measured as the median leverage (debt maturity, debt specialization or debt granularity) of all firms in the same 4-digit industry as the focal firm. Finally, we also measure the *number of employees* in full-time equivalents.

# 4. Results

#### *4.1. Descriptive statistics*

Table 1 presents summary statistics for the entire sample with all firm-year observations and for the startup subsample with first-year observations only. Table 1 shows that firms in the startup sample have higher leverage ratios, relative to firms in the entire sample. The high leverage ratios of startups are in line with recent findings of Robb and Robinson (2014) and Vanacker and Deloof (2015). In addition, firms in the startup sample tend to have a slightly more specialized debt structure, a higher debt maturity and a slightly lower debt granularity, relative to firms in the entire sample. However, these differences largely reflect changes in industry leverage, debt specialization, debt maturity and debt granularity, respectively. Founder-CEO departures are quite common and even in the first year of operation founder-CEOs leave their function in 5.5% of the startups.<sup>9</sup> Founder-CEO deaths are rare events. Unsurprisingly, firms in the startup sample are smaller, exhibit lower profitability, have higher capital expenditures and are less creditworthy, relative to firms in the entire sample.

# \*\*\* Include Table 1 about here \*\*\*

Table 2 shows the correlations between the dependent, independent and control variables, expect for industry and year dummies. The high correlations between the distinct debt policy variables (dependent variables) and their initial values (independent variables) already provide preliminary evidence that initial debt policies are important drivers of future debt policies. Multicollinearity is unlikely to unduly influence our subsequent results as variance inflation factors in all models (unreported) are well below the critical threshold of 10 (Kutner et al., 2004).

<sup>&</sup>lt;sup>9</sup> Founder-CEO departures in the initial year of operation do not reflect bankruptcies in the initial year of operation. The founder-CEO departure variable does not get a value equal to 1 when a firm exits as a consequence of a bankruptcy or another event, because such an approach would mix firm exits with founder-CEO departures.

# 4.2. The influence of initial debt policies on future debt policies

We investigate the influence of initial debt policies on future debt policies by estimating the following ordinary least squares (OLS) regressions<sup>10</sup>:

$$Y_{it} = \alpha + \beta X_{i,t-1} + \gamma Y_{i0} + \nu_t + \varepsilon_{i,t} \quad (5)$$

where  $Y_{it}$  is the debt policy (leverage, debt specialization, debt maturity or debt granularity, respectively) of firm i at time t; *X* is a set of previously identified capital and debt structure determinants that are lagged one year;  $Y_{i0}$  represents a firm's debt policy in the initial year of operation (leverage, debt specialization, debt maturity or debt granularity, respectively); *v* is a time fixed effect, and  $\varepsilon$  is a random error term. Table 3 reports the estimated coefficients and robust standard errors clustered at the firm level.<sup>11</sup>

# \*\*\* Include Table 3 about here \*\*\*

For each dependent variable (leverage, debt specialization, debt maturity and debt granularity), we first estimate models in which the only independent variable is the initial debt policy. In models (a) (leverage), (d) (debt specialization), (g) (debt maturity) and (j) (debt granularity), the effects of initial debt policies on future debt policies are highly statistically significant and also economically significant. A one-standard deviation increase in a startup's initial leverage (debt specialization, debt maturity or debt granularity) corresponds to an average

<sup>&</sup>lt;sup>10</sup> Because the dependent variables are bounded above and below, we also used Tobit regressions as robustness checks. Results remain qualitatively similar.

<sup>&</sup>lt;sup>11</sup> We also perform the regression analyses by clustering both at the firm level and at the year level. By clustering on two dimensions simultaneously, it is possible to capture the unspecified correlation between observations on the same firm in different years and between observations on different firms in the same year completely in cases in which the time effect is not fixed (Petersen, 2009). The regression results remain qualitatively similar.

increase of 10.5% (6.7%, 6.3% or 15.1%) in future values of leverage (debt specialization, debt maturity or debt granularity).

In models (b), (e), (h) and (k), we add year and industry fixed effects and variables that are consistently shown to be important capital and debt structure determinants (e.g., Brav, 2009; Colla et al., 2013). Adding the traditional determinants increases the adjusted R-squared for leverage from 17.7% in model (a) to 43.6% in model (b). While the coefficient of initial leverage becomes smaller when adding these additional variables, initial leverage still remains very important. After adding the traditional debt structure determinants, for debt specialization, the adjusted R-squared increases from 9.8% in model (d) to 12.9% in model (e), for debt maturity, it increases from 10.7% in model (g) to 22.6% in model (h), and for debt granularity, it increases from 19.9% in model (h) to 37.0% in model (k). While the influence of initial debt specialization, initial debt maturity structure and initial debt granularity choices decreases when adding additional variables, they remain economically very important determinants of the future debt specialization, debt maturity structure and debt granularity choices, respectively.

Given our limited understanding of the effects of the traditional capital structure and debt structure variables on the debt policies in new entrepreneurial firms (e.g., Robb and Robinson, 2014), it is also interesting to take a closer look at the coefficients of the control variables. Our findings on the traditional determinants of leverage are consistent with the capital structure literature. In line with Brav (2009) and Rajan and Zingales (1995), we find that firm size and tangibility are positively correlated with leverage, while profitability is negatively correlated with leverage. Consistent with the findings of Brav (2009) for private firms, firm growth is positively correlated with leverage. High creditworthy firms have lower leverage, relative to low creditworthy firms (Vanacker and Deloof, 2015). Finally, industry median leverage is positively correlated with leverage (Lemmon et al., 2008), although the correlation is economically modest.

For the effects of the traditional determinants on debt specialization choices, we find no relation between firm size and debt specialization (Colla et al., 2013). Firm profitability is negatively correlated with debt specialization, while we find a positive correlation between tangibility and debt specialization. The latter finding is in line with Bolton and Scharfstein (1996), who argue that firms with more easily redeployable assets will have a more specialized debt structure. Firm growth, capital expenditures and creditworthiness have an economically modest correlation with debt specialization. Industry median debt specialization is positively correlated with debt specialization.

Turning to the effects of the traditional determinants of debt maturity structure, we find a positive correlation between firm size and debt maturity, which is consistent with the idea that smaller firms issue short-term debt to reduce agency problems (Smith and Warner, 1979). Profitability has a statistically significant negative impact on debt maturity. In line with Morris (1976), who argues that firms try to match the maturity of debt with the maturity of their assets, tangibility positively correlates with debt maturity. Surprisingly, firm growth is positively correlated with debt maturity, which contradicts evidence from more established private firms (Heyman, Deloof and Ooghe, 2008). Capital expenditures are also positively correlated with debt maturity. Creditworthiness and industry median debt maturity have an economically modest correlation with debt maturity.

With respect to debt granularity, we find that larger firms have a tendency to spread out their debt maturity dates more. Profitability correlates negatively with debt granularity, while tangibility, firm growth and capital expenditures are positively correlated with debt granularity. These findings are in line with Choi et al. (2014) for public firms. The medium creditworthiness variable indicates that firms with medium creditworthiness spread out their debt maturity dates

more compared to firms with a low creditworthiness. Industry median debt granularity is positively correlated with debt granularity.

We conduct two additional analyses to test for the robustness of our results. First, the effect of initial debt policies we find in table 3 might reflect an influence of initial values of the traditional capital and debt structure determinants on future debt policies. When this is the case, the influence of initial leverage (initial debt specialization, initial debt maturity and initial debt granularity, respectively) should disappear when the values of the initial traditional determinants are added. However, the results of models (c), (f), (i) and (l) in table 3 show that the initial debt policies remain significant when the initial determinants are added. This suggests that the effects of initial debt policies cannot be explained by the initial values of the traditional capital and debt structure determinants.

Second, we test if the large average impact of initial debt policies on future debt policies (as shown in table 3) is driven by a large influence of initial debt policies during the early years in the firm's life cycle, despite a minimal influence in the later years (e.g., DeAngelo and Roll, 2015). To do this, we estimate equation 5 for a subsample in which we only retain the observations when firms are six years or older and a subsample in which we only retain the observations when firms are 11 years or older.<sup>12</sup> Panel A of table 4 shows that firms' debt policies in their initial year of operation remain statistically and economically significant determinants of future debt policies when we only retain the firm-year observations for which firms are six years or older in panel B, the results remain qualitatively similar. While the influence of initial debt policies becomes smaller as firms age, they remain important determinants of entrepreneurial firms' future debt policies.

<sup>&</sup>lt;sup>12</sup> The coefficients of the control variables are not reported but are in line with those in table 3.

In sum, our results show the existence of an important stable component in entrepreneurial firms' debt policies. Indeed, initial debt policies, which are time-invariant factors, are one of the most important drivers of future debt policies even when we control for traditional contemporaneous determinants.

# 4.3. The importance of firm-specific effects on debt policies

Next, we analyze the importance of time-invariant, firm-specific factor(s) by conducting a variance decomposition of the debt policies. Specifically, we use the following equation:

$$Y_{it} = \alpha + \beta X_{i,t-1} + \eta_i + \nu_t + \varepsilon_{i,t} \quad (6)$$

where  $\eta$  *represents* the firm fixed effect in the equation and all other variables as defined in equation (5).

Table 5 reports the fraction of the total partial sum of squares of the respective model captured by each variable or effect. Panel A represents the results of the variance decomposition of leverage, while panels B, C and D report the results of the variance decomposition of debt specialization, debt maturity and debt granularity, respectively.

# \*\*\* Include Table 5 about here \*\*\*

Models (a) of each panel, which include only firm fixed effects, explains 56.8% (40.9%, 42.9% and 56.6%) of the total variation in leverage (debt specialization, debt maturity and debt granularity) of our sample. Models (b) show that the industry fixed effects do not explain much of the total variation in leverage (debt specialization, debt maturity and debt granularity). Similarly, the year fixed effects do not explain much of the total variation in the models as

depicted in models (c) and (d). These findings suggest that time-invariant factors account for the majority of variation in capital structures of new entrepreneurial firms. It also suggests that theories based on time-varying factors can offer only a rather incomplete explanation for the heterogeneity in capital and debt structures in a time-series study.

Model (e) shows the results of equation (6) when using the traditional capital and debt structure determinants as previously specified. These variables are able to explain 30.6% (5.8%, 18.4% and 28.5%) of the leverage (debt specialization, debt maturity and debt granularity) variation, which is much lower than the explanatory power of models (a), which simply include firm fixed effects.

Adding firm fixed effects leads to a large increase in the adjusted  $R^2$  from model (e) to model (f) for all debt variables. For leverage, it increases from 30.6% to 71.2%, for debt specialization from 5.8% to 44.2%, for debt maturity from 18.4% to 51.8%, and for debt granularity from 28.5% to 64.3%.

In sum, the results of the variance decompositions suggest that there is an important unobserved firm-specific factor that drives the debt policies of new entrepreneurial firms. This observation was also made by Lemmon et al. (2008) for leverage decisions, but our results suggest that this unobserved factor is already present at startup and affects a broader range of debt policies. This unobserved factor cannot be captured with traditional capital structure and debt structure variables.<sup>13</sup> Second, most of the variation in the debt policies stems from cross-sectional differences, as opposed to within-firm or time-series variation.

<sup>&</sup>lt;sup>13</sup> In line with DeAngelo and Roll (2015), we also run additional models in which we include firm-time interactions. Although part of the explanatory power attributed to firm fixed effects is due to suppression of these interaction effects, firm fixed effects remain very important.

### 4.4. The influence of founder-CEO departures on the evolution of debt policies

Our findings so far raise the question to what extent the stable component of entrepreneurial firms' debt policies is determined by the founder-CEO. To investigate the impact of the founder-CEO on debt stability, we estimate firm-fixed-effects regressions, which allow us to control for (stable) unobserved firm-specific factors. This is important because there might be other stable unobserved firm-specific factors in addition to the founder-CEO that drive entrepreneurial firms' debt policies. Specifically, we estimate the following regression equation:

$$Y_{it} = \alpha + \beta X_{i,t-1} + \gamma * \text{Founder} - \text{CEO Departure} + w(Y_{i0} * \text{Founder} - \text{CEO Departure}) + \nu_t + \varepsilon_{i,t} (7)$$

where  $Y_{it}$  is the debt policy (i.e., leverage, debt specialization, debt maturity or debt granularity) of firm i at time t; *X* is a set of previously identified capital and debt structure determinants that are lagged one year; *Founder-CEO Departure* is a dummy that equals 1 from the year of the founder-CEO departure onward and 0 otherwise;  $Y_{i0} * Founder-CEO Departure$  represents the interaction between firm's initial debt policy (i.e., initial leverage, debt specialization, debt maturity or debt granularity) and *Founder-CEO Departure*;<sup>14</sup> *v* is a time fixed effect, and  $\varepsilon$  is a random error term. Table 6 reports the estimated coefficients and robust standard errors clustered at the firm level. All models include firm and year fixed effects.

# \*\*\* Include Table 6 about here \*\*\*

Models (a), (b), (c) and (d) of table 6 indicate that founder-CEO departures have a statistically significant positive impact on leverage, debt specialization and debt maturity, while they have a statistically significant negative impact on debt granularity. Hence, founder-CEO

<sup>&</sup>lt;sup>14</sup> Note that the main effects of the initial debt policies are now absorbed in the firm fixed effects, which also control for any other unmeasured but stable firm characteristic that influences firms' debt policies.

successors use more debt financing, particularly long-term debt financing, and have a less diversified debt structure (i.e., they use less debt sources and their debt maturity dates are less spread out). These findings are in line with the view that founder-CEOs can be relatively conservative in their financial decision-making (e.g., Ang, 1991).

However, for the purpose of our study, we are primarily interested in the interaction between initial debt policies and founder-CEO departures. We expect this interaction, i.e. the coefficient w from equation 7, to be significantly negative. When the founder-CEO departs, the influence of the initial debt policies on the future debt policies significantly declines, relative to when the founder-CEO remains in function. Consistent with our expectation, models (a), (b), (c) and (d) of table 6 show that after the departure of the founder-CEO, the influence of a firm's initial leverage (debt specialization, debt maturity and debt granularity) on future values of leverage (debt specialization, debt maturity and debt granularity) is 9.3% (3.9%, 3.3% and 5.7%) lower.<sup>15</sup> These findings provide support for the view that founder-CEOs "imprint their mark" on the debt policies of the firms they manage; as founder-CEOs leave their function the impact of initial debt policies on future debt policies decreases significantly.

As Fee et al. (2013) note, the endogenous nature of CEO departures makes it difficult to determine whether policy changes after a change in a firm's CEO are caused by the departure of the CEO, by a decision of the firm's board or by another unobserved variable.<sup>16</sup> Therefore, we replace the founder-CEO departure dummy in equation 7 by a founder-CEO death dummy. Focusing on founder-CEO deaths allows us to examine the impact of exogenous founder-CEO

<sup>&</sup>lt;sup>15</sup> When founder-CEOs depart during the earliest phase in the firm's existence, they may not have had sufficient time to imprint their mark, or alternatively, their policies may not have been institutionalized yet. Consistent with this idea, unreported supplementary regressions confirm that founder-CEO departures more strongly decrease the effect of initial debt policies on future debt policies when founder-CEO departures happen during the first five (six or seven) years after founding relative to when they happen after the first five (six or seven) years after founding.

<sup>&</sup>lt;sup>16</sup> In our research context, it is unlikely that the policy changes after a change in a firm's CEO are caused by a decision by a firm's board because most of the firms in our sample do not have a board.

departures on the evolution of entrepreneurial firms' debt policies. In this way, it is possible to investigate whether the decrease in importance of initial debt policies on entrepreneurial firms' future debt policies after the departure of the founder-CEO is caused by CEO-style effects. If the style of founder-CEOs has an impact on the debt policies of new entrepreneurial firms, we would expect to find a decline in the importance of initial debt policies after exogenous founder-CEO departures.

# \*\*\* Include Table 7 about here \*\*\*

The results in table 7 using founder-CEO deaths confirm our results from table 6. Founder-CEO deaths increase firms' leverage, debt specialization (although not statistically significant) and debt maturity ratios, while they lower the extent to which firms spread out their debt maturity dates across time. More importantly, we find in table 7 that after the founder-CEO dies, the impact of initial leverage, debt maturity and debt granularity on future leverage, debt maturity and debt granularity declines. The impact of initial debt specialization on future debt specialization also declines after exogenous founder-CEO departures, but this effect is not significant at traditional levels (p = 0.111). These results suggest that founder-CEOs imprint their mark on the debt policies of their firms, causing a stable component in entrepreneurial firms' debt policies. However, firms change inertial debt policies after founder-CEO departures.<sup>17</sup>

In sum, the regression analyses in this section show that entrepreneurial firms' initial debt policies become less impactful for their future debt policies after the departure of the founder-CEO. These results suggest that one unobserved firm-specific factor that drives the debt policy decisions of new entrepreneurial firms is the founder-CEO.

<sup>&</sup>lt;sup>17</sup> There are only 19 firms in our sample where the founder-CEO dies in the first 15 years after startup. This small number might lead to inefficiency in the estimation and unreliable point estimates. However, despite these problems, our results for founder-CEO death and its interaction with initial debt policy is consistent with the founder-CEO departure analyses.

### 5. Alternative explanations and robustness checks

We interpret our findings in the context of imprinting theory, which implies that founding decisions play an important role in imprinting firm characteristics that are perpetuated over time. However, there are some alternative explanations for our findings, which we discuss below. Detailed results of the robustness checks discussed below are available in the Internet Appendix.

First, it is possible that the observed stability of debt policies at the firm level is driven in large part by low speeds of adjustment to moving target debt policies. To investigate this possibility, we run partial-adjustment models of firm debt policies and estimate the speed of adjustment of firms' actual debt policies to their target debt policies (see Lemmon et al. (2008) for a similar approach). However, results are not consistent with this alternative explanation. Specifically, partial-adjustment models that include firm-fixed effects exhibit higher model fit and higher speeds of adjustment, relative to models without firm-fixed effects. Hence, time-invariant, firm-specific factors drive target debt policies. Overall, our findings suggest that not only entrepreneurial firms' debt policies but also their target debt policies contain an important stable component.

A second alternative explanation for our findings is that it is the presence of the founder-CEO in the firm rather than his/her preference for a particular debt policy that matters. A CEO change may have a large impact on the firm's fundamental characteristics and access to external finance, and as a result affect its capital structure. Especially the prospects of new and small firms, like the ones we study, are often closely tied to founder-CEOs' skills (Cooper, Gimeno and Woo, 1994), which implies that founder-CEO departures (deaths) may have a strong negative impact on the firm's future prospects. Furthermore, discontinued lending relationships after founder-CEOs departures (deaths) might increase adverse selection and moral hazard risks between firms and their lenders (Berger and Udell, 1995; Petersen and Rajan, 1994). If it is the simple presence of founder-CEOs in firms that matters, we therefore expect to find that founder-CEO departures (deaths) decrease firm performance and increase the likelihood of going bankrupt. However, we find that firm profitability increases after the departure of the founder-CEO, although the effect is economically very small. Founder-CEO deaths do not influence subsequent firm profitability. In addition, founder-CEO departures do not have a significant impact on the likelihood of going bankrupt, while founder-CEO deaths decrease the likelihood of going bankrupt. Thus, these additional tests contradict the argument that the simple presence of founder-CEOs is more important than founder-CEOs' preferences for a particular debt policy.

Third, firms characterized by higher levels of information asymmetry may not have access to debt financing and thus initially rely more on equity financing (e.g., Berger and Udell, 1998). As a result, the stability of debt policies might be stronger for firms characterized by lower levels of information asymmetry, while the debt policies of firms characterized by higher levels of information asymmetry might be more in line with the traditional information-based theories on the evolution of entrepreneurial financing. To test for this possibility, we distinguish between firms based on the level of information asymmetry. Specifically, we run separate regressions for startups founded in high-tech industries and startups founded in other industries. However, we fail to find significant differences in the relation between initial debt policies and future debt policies for both subsamples. These findings indicate that differences in the level of information asymmetry do not drive the stable component of firms' debt policies.

Fourth, we test for the possibility that the stable component of entrepreneurial firms' debt policies is caused by firms that have already achieved their desired structure (or size) at startup and, as a consequence, do not grow much in the next 15 years after startup. For this purpose, we compute the average yearly growth rate in total assets of each firm based on the entire period it is in the sample. Based on their average yearly growth rate in total assets, firms are sorted into four

portfolios, i.e. firms with a low, medium, high and very high average yearly growth rate, respectively. We find that initial leverage, debt specialization and debt granularity exert less influence on future leverage, debt specialization and debt granularity for the firms with very high average yearly growth rates. However, even in the sample of firms with a very high average yearly growth rate, the initial debt policies remain an important determinant of future debt policies. Thus, irrespective of firm growth, initial debt policies significantly influence future debt policies.

Fifth, we investigate whether our results are potentially driven by very small firms that may have limited operational activities but dominate the population of entrepreneurial firms (and our sample). Indeed, our sample firms employ on average about 6 employees and the median firm employs 3 people (all in full time equivalents). We analyze subsamples of firms with more than one (five and ten) employee(s) in the year of startup. The results based on these different subsamples remain quantitatively similar. Hence, the stable component of firms' debt policies and the results of the founder-CEO departure analyses are not driven by the smallest firms.

Finally, firms may leave the sample early either due to bankruptcies, acquisitions or buyouts. This may bias our results. To address this issue, we examine the subsample of survivors, i.e., the subsample of firms that are active in all sample years. The unreported results reveal that entrepreneurial firms' debt policies still contain a stable component when we limit the sample to the firms that are active in all sample years.

## 6. Conclusion

This paper is the first to provide evidence on the evolution of a broad range of debt policies, including leverage, debt specialization, debt maturity and debt granularity policies, in entrepreneurial firms from startup. Our analysis is based on a unique dataset, based on the

universe of Belgian firms founded between 1996 and 1998, which we track for up to 15 years after startup.

We find that entrepreneurial firms' leverage, debt specialization, debt maturity and debt granularity policies contain an important time-invariant, stable component that remains present in the 15 years after startup. Specifically, financing decisions at startup serve as strong predictors of future financing decisions, and this is the case even after controlling for the traditional contemporaneous capital structure and debt structure variables, such as firm size, profitability, tangibility and growth. This finding is in line with imprinting theory. Our findings further suggest that current capital structure and debt structure research is missing an important time-invariant, firm-specific factor(s) present from startup that drives the stable component of debt policies. Our results suggest that one important time-invariant, firm-specific factor is the founder-CEO. The influence of initial debt policies of entrepreneurial firms on future debt policies declines significantly after the departure (and death) of the founder-CEO.

Our findings underscore the need for more research on financial decision making in very early stage firms that goes beyond the traditional capital structure and debt structure variables. We provide a fresh perspective to increase our understanding of financial decision making in early stage entrepreneurial firms—i.e., imprinting theory—and hope our paper will encourage others to study more fully the evolution of entrepreneurial firms' debt policies from startup.

#### References

Ang, J. S. (1991). Small business uniqueness and the theory of financial management. The Journal of Entrepreneurial Finance, 1(1), 11-13.

Berger, A. N., & Udell, G. F. (1995). Relationship lending and lines of credit in small firm finance. Journal of Business, 351-381.

Berger, A., Udell, G., 1998. The economics of small business finance: the roles of private equity and debt markets in the financial growth cycle. Journal of Banking and Finance 22(6), 613-673.

Bertrand, M., Schoar, A, 2003. Managing with style: the effects of managers on firm policies. Quarterly Journal of Economics 118(4), 1169-1208.

Boeker, W., 1989. Strategic change: the effects of founding and history. Academy of Management Journal 32(3), 489-515.

Bolton, P., Scharfstein, D.S., 1996. Optimal debt structure and the number of creditors. Journal of Political Economy 104(1), 1-25.

Brav, O., 2009. Access to capital, capital structure and the funding of the firm. Journal of Finance 64(1), 263-308.

Cassar, G., 2004. The financing of business startups. Journal of Business Venturing 19(2), 261-283.

Choi, J., Hackbarth, D., Zechner, J., 2014. Granularity of corporate debt. Working Paper.

Colla, P., Ippolito, F., Li, K., 2013. Debt specialization. Journal of Finance 68(5), 2117-2141.

Cooper, A. C., Gimeno, F. J., & Woo, C. Y. (1994). Initial human and financial capital as predictors of new venture performance. Journal of Business Venturing, 9(5), 371-395.

Cosh, A., Cumming, D., Hughes, A., 2009. Outside entrepreneurial capital. Economic Journal 119(540), 1494 – 1533.

Cumming, D., 2005. Capital structure in venture finance. Journal of Corporate Finance 11(3), 550-585.

Custodio, C., Ferreira, M.A., Laureano, L., 2013. Why are US firms using more short-term debt? Journal of Financial Economics 108(1), 182-212.

DeAngelo, H., Roll, R., 2015. How stable are corporate capital structures? Journal of Finance 70(1), 373-418.

Degryse, H., Masschelein, N., Mitchell, J., 2011. Staying, dropping, or switching: the impacts of bank mergers on small firms. Review of Financial Studies 24(4), 1102-1140.

Demirgüç-Kunt, A., Levine, R., 1999. Bank-based and market-based financial systems: Crosscountry comparisons (Vol. 2143). World Bank Publications.

Dennis, D.J., 2012. The persistent puzzle of corporate capital structure: current challenges and new directions. Financial Review 47(4), 631-643.

Fama, E.F., Jensen, M.C., 1983. Separation of ownership and control. Journal of Law and Economics 26(2), 301-325.

Fee, C.E., Hadlock, C.J., Pierce, J.R., 2013. Managers with and without style: evidence using exogenous variation. Review of Financial Studies 26(3), 567-601.

Groh, A.P., von Liechtenstein, H., Lieser, K., 2010. The European venture capital and private equity country attractiveness indices. Journal of Corporate Finance 16(2), 205-224.

Hanousek, J., Shamshur, A., 2011. A stubborn persistence: is the stability of leverage ratios determined by the stability of the economy? Journal of Corporate Finance 17(5), 1360-1376.

Heyman, D., Deloof, M., & Ooghe, H. (2008). The financial structure of private held Belgian firms. Small Business Economics, 30(3), 301-313.

Huyghebaert, N., Van de Gucht, L. M., 2007. The determinants of financial structure: New insights from business start-ups. European Financial Management 13(1), 101-133.

Kutner, M. H., Nachtsheim, C., & Neter, J. (2004). Applied linear regression models. McGraw-Hill/Irwin.

Lambrecht, B.M., Myers, S.C., 2014. The dynamics of investment, payout and debt. Working paper.

Lemmon, M.L., Roberts, M.R., Zender, J.F., 2008. Back to the beginning: persistence and the cross-section of corporate capital structure. Journal of Finance 63(4), 1575-1608.

Malmendier, U., Tate, G., Yan, J., 2011. Overconfidence and early-life experiences: the effect of managerial traits on corporate financial policies. Journal of Finance 66(5), 1687-1733.

Morellec, E., Nikolov, B., Schürhoff, N., 2012. Corporate governance and capital structure dynamics. Journal of Finance 67(3), 803-848.

Morris, J.R., 1976. On corporate debt maturity strategies. Journal of Finance 31(1), 29-37.

Panier, F., Pérez-González, F., Villanueva, P., 2013. Capital structure and taxes: what happens when you (also) subsidize equity? Working paper

Petersen, M.A., Rajan, R.G., 1994. The benefits of lending relationships: evidence from small business data. Journal of Finance 49(1), 3-37.

Petersen, M.A., 2009. Estimating standard errors in finance panel data sets: comparing approaches. Review of Financial Studies 22(1), 435-480.

28

Rajan, R.G., Zingales, L., 1995. What do we know about capital structure? Some evidence from international data. Journal of Finance 50(5), 1421-1460.

Robinson, D., 2012. New perspectives on entrepreneurial capital structure. In D. Cumming (ed.), Oxford Handbook of Entrepreneurial Finance 153-166.

Robb, A.M., Robinson, D.T., 2014. The capital structure decisions of new firms. Review of Financial Studies 27(9), 153-179.

Rotemberg, J.J., Saloner, G., 2000. Visionaries, managers and strategic direction. RAND Journal of Economics 31(4), 693-716.

Scherr, F.C., Hulburt, H.M., 2001. The debt maturity structure of small firms. Financial Management 30(1), 85-111.

Schoar, A., Zuo, L., 2014. Shaped by booms and busts: how the economy impacts CEO careers and management styles. Working Paper.

Smith, C.W. Jr., Warner, J.B., 1979. On financial contracting. Journal of Financial Economics 7(2), 117-161.

Stinchcombe, A., 1965. Social structures and organizations. In J.G. March (ed.), Handbook of organizations. 142-193.

Titman, S., Wessels, R., 1988. The determinants of capital structure choice. Journal of Finance 43(3), 1-19.

Vanacker, T., Deloof, M., 2015. The financial and real effects of credit availability for startup firms: evidence from the recent financial crisis. Working paper.

Welch, I., 2004. Capital structure and stock returns. Journal of Political Economy 112(1), 106-131.

Wu, X., Yeung, C.K.A., 2012. Firm growth type and capital structure persistence. Journal of Banking and Finance 36(12), 3427-3443.

# Appendix A. Variable definitions

Variables	Definitions
Leverage	total debt/total assets.
Debt Specialization	Herfindahl-Hirschman Index of debt usage that refers to the degree of debt specialization. HHI is equal to one when a firm only uses one type of debt, while HHI equals zero when a firm simultaneously uses all types of debt in equal proportion.
Debt Maturity	the percentage of total debt that matures in more than five years.
Debt Granularity	Herfindahl-Hirschman Index of debt maturity that refers to the degree to which a firm spreads out its debt maturity dates across time.
	HHI is equal to one when a firm does not spread out its debt maturity dates across time, while HHI equals zero when a firm completely
	spreads out its debt maturity dates across time. This measure is multiplied by minus one so that higher values of debt granularity
	indicate that firms increasingly spread out their debt maturity dates.
Initial Leverage (Debt Specialization, Debt	the value of leverage (debt specialization, debt maturity and debt granularity) at startup.
Maturity and Debt Granularity,)	
Founder-CEO Departure	a dummy variable that equals 1 from the year of the founder-CEO departure onward and 0 otherwise.
Founder-CEO Death	a dummy variable that equals 1 from the year of the founder-CEO death onward and 0 otherwise.
Firm Size	ln(total assets).
Profitability	EBIT/total assets.
Tangibility	net PPE/total assets.

Firm Growth	(total assets in year t – total assets in year t-1)/total assets in year in year t-1
Capital Expenditures	new investments in fixed assets/total assets.
Dummy Low (Medium and High)	a dummy variable that equals 1 for firms with a low (medium and high) creditworthiness and 0 otherwise. Firms are classified into three
Creditworthiness	categories according to their FiTo score (i.e., a default risk indicator from Graydon-the market leader in commercial and marketing
	information as well as credit and debt management in Belgium). Firms with a low (medium and high) creditworthiness were taken to be
	those situated below the 25th percentile (those situated between the 25th and the 75th percentile; and those situated above the 75th
	percentile).
Industry Median Leverage	the median leverage (debt specialization, debt maturity and debt granularity) of all firms in the same industry as the focal firm.
(Debt Specialization,	
Debt Maturity and Debt Granularity)	

#### **Table 1: Descriptive statistics**

The sample consists of 4,962 Belgian firms founded between 1996 and 1998 that are followed for up to 15 years after startup. The table presents the mean, median (in brackets) and standard deviations (SD) for the entire sample of all firm-year observations and the subsample of new incorporations. Variable definitions are provided in the appendix.

	All firm	-year	First-y	ear
	observa	tions	observa	tions
Variable	Mean	SD	Mean	SD
	[Median]		[Median]	
Leverage	0.629	0.249	0.753	0.199
-	[0.682]		[0.808]	
Debt Specialization	0.401	0.212	0.419	0.212
-	[0.356]		[0.373]	
Debt Maturity	0.096	0.203	0.128	0.269
	[0.000]		[0.000]	
Debt Granularity	-0.534	0.338	-0.555	0.340
-	[-0.452]		[-0.476]	
Founder-CEO Departure	0.273	_	0.055	_
-	_			
Founder-CEO Death	0.002	_	0.000	_
	_			
Firm Size	5.838	1.416	4.884	1.636
	[5.877]		[5.050]	
Profitability	0.063	0.108	0.049	0.121
	[0.043]		[0.025]	
Tangibility	0.303	0.255	0.322	0.258
	[0.240]		[0.265]	
Firm Growth	0.071	0.262		_
	[0.010]			
Capital Expenditures	0.025	0.040	0.084	0.032
	[0.000]		[0.100]	
Dummy Low Creditworthiness	0.247	_	0.360	_
	_			
Dummy Medium Creditworthiness	0.485	_	0.533	
	—			
Dummy High Creditworthiness	0.269		0.107	
	—		_	
Industry Median Leverage	0.668	0.085	0.724	0.080
	[0.664]		[0.724]	
Industry Median Debt Specialization	0.355	0.052	0.362	0.055
	[0.344]		[0.353]	
Industry Median Debt Maturity	0.005	0.047	0.015	0.080
	[0.000]		[0.000]	
Industry Median Debt Granularity	-0.612	0.162	-0.613	0.157
	[-0.596]		[-0.599]	
Number of Employees	6.297	13.345	3.749	5.352
	[3.000]		[2.000]	
Number of Observations <sup>18</sup>	49,418		4,962	

<sup>&</sup>lt;sup>18</sup> The number of observations for debt specialization, debt maturity, debt granularity, industry median debt specialization, industry median debt maturity and industry median debt granularity equals 48,185 (all firm-year observations) and 4,853 (first-year observations), respectively.

# Table 2: Correlation matrix

The sample consists of 4,962 Belgian firms founded between 1996 and 1998 that are followed for up to 15 years after startup. All correlations with an absolute value equal or higher than 0.010 are statistically significant at the 5% significance level.

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)
(1)	Leverage	1.000																				
(2)	Debt Specialization	-0.047	1.000																			
(3)	Debt Maturity	0.214	0.193	1.000																		
(4)	Debt Granularity	0.298	-0.257	0.476	1.000																	
(5)	Initial Leverage	0.419	-0.053	0.088	0.184	1.000																
(6)	Initial Debt Specialization	0.012	0.312	0.076	-0.075	0.055	1.000															
(7)	Initial Debt Maturity	0.109	0.058	0.327	0.281	0.235	0.144	1.000														
(8)	Initial Debt Granularity	0.111	-0.093	0.203	0.446	0.297	-0.215	0.497	1.000													
(9)	Founder-CEO Departure	-0.090	0.019	-0.024	-0.046	-0.045	0.009	-0.020	-0.042	1.000												
(10)	Founder-CEO Death	-0.009	0.001	0.003	0.003	-0.003	-0.006	0.008	0.000	0.082	1.000											
(11)	Firm Size	0.098	0.037	0.159	0.122	0.140	0.098	0.171	0.073	0.149	0.003	1.000										
(12)	Profitability	-0.294	-0.116	-0.106	-0.097	-0.040	-0.016	-0.046	-0.034	0.003	0.006	0.075	1.000									
(13)	Tangibility	0.249	0.093	0.383	0.498	0.089	0.070	0.234	0.291	-0.024	-0.019	0.055	-0.173	1.000								
(14)	Firm Growth	0.129	-0.004	0.040	-0.004	-0.053	-0.024	-0.070	-0.067	-0.022	-0.004	0.105	0.205	-0.019	1.000							
(15)	Capital Expenditures	0.152	-0.008	0.079	0.114	-0.009	-0.029	-0.026	0.028	-0.087	-0.012	-0.075	-0.028	0.188	0.369	1.000						
(16)	Dummy Low Creditworthiness	0.148	0.072	0.026	-0.005	-0.039	-0.006	-0.007	-0.007	0.011	-0.007	-0.172	-0.619	0.094	-0.148	-0.001	1.000					
(17)	Dummy Medium Creditworthiness	0.242	-0.013	0.087	0.141	0.128	0.011	0.069	0.071	-0.045	0.004	0.047	0.007	0.109	0.064	0.034	-0.528	1.000				
(18)	Dummy High Creditworthiness	-0.406	-0.055	-0.135	-0.156	-0.106	-0.006	-0.070	-0.073	0.055	0.005	0.125	0.602	-0.217	0.074	-0.080	-0.347	-0.588	1.000			
(19)	Industry Median Leverage	0.149	0.005	0.042	0.017	0.070	0.015	0.017	0.029	-0.048	-0.024	-0.116	-0.078	0.086	0.010	0.091	0.056	0.060	-0.148	1.000		
(20)	Industry Median Debt Specialization	-0.037	0.168	0.046	-0.071	-0.055	0.162	0.034	-0.075	0.027	-0.019	0.090	0.001	0.062	0.001	0.002	0.011	-0.035	0.021	0.001	1.000	
(21)	Industry Median Debt Maturity	0.029	0.030	0.059	0.019	0.006	0.042	0.041	0.019	-0.029	-0.005	0.015	-0.017	0.057	0.003	0.038	0.014	0.006	-0.029	0.055	0.065	1.000
(22)	Industry Median Debt Granularity	-0.094	-0.041	-0.009	0.132	0.090	-0.026	0.077	0.172	0.104	0.013	0.120	0.002	0.130	-0.102	-0.174	-0.032	-0.051	0.073	-0.140	-0.110	0.007

#### Table 3: The influence of initial debt policies

The sample consists of 4,962 Belgian firms founded between 1996 and 1998 that are followed for up to 15 years after startup. Coefficients in all specifications are estimated using OLS. To facilitate the comparison of the coefficients, we standardize the independent and the control variables to have zero mean and unit variance. Standard errors are reported in parentheses and are computed robust to both clustering at the firm level and heteroskedasticity. Variable definitions are provided in the appendix. \*\*\*, \*\*, and \* denote statistical significance at the 1%; 5% and 10% level, respectively.

Dependent Variable: Debt Policy		Leverage		De	bt Specializat	ion	]	Debt Maturity	y	D	ebt Granulari	ty
	(a)	(b)	(c)	(d)	(e)	(f)	(g)	(h)	(i)	(j)	(k)	(1)
Initial Debt Policy	0.105***	0.087***	0.093***	0.067***	0.060***	0.061***	0.063***	0.044***	0.052***	0.151***	0.104***	0.117***
	(0.003)	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)	(0.003)	(0.003)	(0.003)	(0.004)	(0.004)	(0.004)
Firm Size		0.041***	0.066***		0.000	0.006		0.024***	0.045***		0.033***	0.064***
		(0.003)	(0.004)		(0.003)	(0.004)		(0.002)	(0.004)		(0.003)	(0.006)
Profitability		-0.028***	-0.028***		-0.021***	-0.023***		-0.004***	-0.007***		-0.006**	-0.009***
		(0.002)	(0.002)		(0.002)	(0.002)		(0.002)	(0.002)		(0.003)	(0.003)
Tangibility		0.028***	0.047***		0.013***	0.017***		0.058***	0.069***		0.126***	0.148***
		(0.002)	(0.003)		(0.003)	(0.003)		(0.003)	(0.003)		(0.004)	(0.004)
Firm Growth		0.043***	0.040***		0.003***	0.003**		0.016***	0.016***		0.016***	0.015***
		(0.001)	(0.001)		(0.001)	(0.001)		(0.001)	(0.001)		(0.001)	(0.002)
Capital Expenditures		0.011***	0.006***		-0.004***	-0.005***		0.004***	0.001		0.019***	0.013***
		(0.001)	(0.001)		(0.001)	(0.001)		(0.001)	(0.001)		(0.001)	(0.001)
Dummy Medium Creditworthiness		0.006***	0.006***		-0.002	-0.001		0.001	0.001		0.023***	0.021***
		(0.002)	(0.002)		(0.002)	(0.002)		(0.002)	(0.002)		(0.003)	(0.003)
Dummy High Creditworthiness		-0.062***	-0.059***		0.007**	0.006**		-0.006***	-0.006***		0.006*	-0.005
		(0.003)	(0.003)		(0.003)	(0.003)		(0.002)	(0.002)		(0.004)	(0.004)
Industry Median Debt Policy		0.005**	0.003*		0.017***	0.015***		0.004**	0.003*		0.020***	0.023***
		(0.002)	(0.002)		(0.003)	(0.003)		(0.002)	(0.002)		(0.004)	(0.004)
Initial Firm Size			-0.038***			-0.007			-0.029***			-0.042***
			(0.004)			(0.004)			(0.004)			(0.006)
Initial Profitability			-0.012***			0.009**			0.005*			0.004
-			(0.004)			(0.004)			(0.003)			(0.005)
			()									(/

Initial Tangibility			-0.033***			-0.005			-0.021***		-0.044***		
			(0.003)			(0.003)			(0.003)			(0.005)	
Initial Firm Growth		0.011***				0.000			0.000	0.001			
			(0.003)		(0.002)				(0.002)		(0.003)		
Initial Capital Expenditures		0.000				-0.006**		0.001			0.014**		
		(0.002)				(0.002)			(0.002)			(0.003)	
Initial Dummy Medium			0.000			-0.005			0.002			0.007*	
Creditworthiness			(0.003)			(0.003)			(0.003)			(0.004)	
Initial Dummy High			0.003			0.000			0.002			-0.003	
Creditworthiness			(0.004)			(0.004)			(0.003)			(0.005)	
Initial Industry Median			0.004			0.003			0.003			-0.008**	
Debt Policy			(0.003)			(0.004)			(0.003)			(0.004)	
Year FE	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	
Industry FE	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	
Adjusted R <sup>2</sup>	0.177	0.436	0.460	0.098	0.129	0.133	0.107	0.226	0.244	0.199	0.370	0.388	
Observations	44,456	44,456	44,456	43,332	43,332	43,332	43,332	43,332	43,332	43,332	43,332	43,332	

### Table 4: The influence of initial debt policies based on subsamples

The sample consists of 4,962 Belgian firms founded between 1996 and 1998. Coefficients in all specifications are estimated using OLS. To facilitate the comparison of the coefficients, we standardize the independent and the control variables to have zero mean and unit variance. Standard errors are reported in parentheses and are computed robust to both clustering at the firm level and heteroskedasticity. The control variables included in the models, which are not reported in this table, correspond with those included in the corresponding models in table 3. Variable definitions are provided in the appendix. \*\*\*, \*\*, and \* denote statistical significance at the 1%; 5% and 10% level, respectively.

Dependent Variable		Leverage		Deb	ot Specializa	tion	Ι	Debt Maturit	у	De	ebt Granular	ity
	(a)	(b)	(c)	(d)	(e)	(f)	(g)	(h)	(i)	(j)	(k)	(1)
Initial Debt Policy	0.086***	0.067***	0.071***	0.047***	0.039***	0.040***	0.038***	0.018***	0.026***	0.120***	0.073***	0.082***
	(0.004)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.003)	(0.005)	(0.004)	(0.005)
Controls	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Year FE	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Industry FE	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Adjusted R <sup>2</sup>	0.114	0.367	0.390	0.047	0.093	0.096	0.044	0.199	0.216	0.124	0.347	0.361
Observations	28,401	28,401	28,401	27,697	27,697	27,697	27,697	27,697	27,697	27,697	27,697	27,697

#### Panel A: Subsample with observations when firms are six years or older

#### Panel B: Subsample with observations when firms are 11 years or older

Dependent Variable		Leverage		Deb	ot Specializa	tion	Ι	Debt Maturit	у	De	ebt Granular	ity
	(a)	(b)	(c)	(d)	(e)	(f)	(g)	(h)	(i)	(j)	(k)	(1)
Initial Debt Policy	0.077***	0.060***	0.061***	0.040***	0.033***	0.034***	0.028***	0.010***	0.015***	0.102***	0.058***	0.065***
	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.004)	(0.003)	(0.004)	(0.006)	(0.005)	(0.006)
Controls	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Year FE	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Industry FE	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes	No	Yes	Yes
Adjusted R <sup>2</sup>	0.086	0.295	0.318	0.033	0.083	0.086	0.023	0.197	0.208	0.088	0.335	0.348
Observations	12,364	12,364	12,364	12,097	12,097	12,097	12,097	12,097	12,097	12,097	12,097	12,097

#### Table 5: Variance decomposition of debt policies

The sample consists of 4,962 Belgian firms founded between 1996 and 1998 that are followed for up to 15 years after startup. Variable definitions are provided in the appendix. Firm FE are firm fixed effects. Year FE are year fixed effects. Industry FE are industry fixed effects. Panels A, B, C and D present the results of the variance decomposition of leverage, debt specialization, debt maturity and debt granularity, respectively.

			L	everage		
Variable	(a)	(b)	(c)	(d)	(e)	(f)
Firm FE	1.000		•	0.895		0.934
Year FE			1.000	0.105	0.310	0.048
Firm Size					0.229	0.007
Profitability					0.035	0.002
Tangibility					0.106	0.006
Firm Growth					0.045	0.002
Capital Expenditures					0.000	0.000
Dummy Medium Creditworthiness					0.007	0.000
Dummy High Creditworthiness					0.125	0.001
Industry Median Leverage					0.006	0.000
Industry FE		1.000			0.138	
Adj. R <sup>2</sup>	0.568	0.043	0.093	0.645	0.306	0.712

#### **Panel A: Leverage**

#### Panel B: Debt Specialization

			Debt Sp	pecializatio	on	
Variable	(a)	(b)	(c)	(d)	(e)	(f)
Firm FE	1.000			0.998		0.993
Year FE			1.000	0.002	0.017	0.001
Firm Size					0.003	0.000
Profitability					0.156	0.004
Tangibility					0.126	0.000
Firm Growth					0.010	0.000
Capital Expenditures					0.035	0.000
Dummy Medium Creditworthiness					0.000	0.000
Dummy High Creditworthiness					0.015	0.000
Industry Median Debt Specialization					0.149	0.000
Industry FE		1.000	•		0.488	•
	0.400	0.025	0.001	0.410	0.059	0.442
Auj. K <sup>2</sup>	0.409	0.035	0.001	0.410	0.058	0.442

# Panel C: Debt Maturity

			Deb	t Maturity		
Variable	(a)	(b)	(c)	(d)	(e)	(f)
Firm FE	1.000			0.975		0.955
Year FE	•		1.000	0.025	0.020	0.010
Firm Size					0.097	0.010
Profitability					0.001	0.000
Tangibility					0.721	0.022
Firm Growth					0.021	0.002
Capital Expenditures					0.002	0.000
Dummy Medium Creditworthiness					0.001	0.000
Dummy High Creditworthiness					0.002	0.000
Industry Median Debt Maturity					0.001	0.000
Industry FE		1.000			0.135	
Adj. R <sup>2</sup>	0.429	0.025	0.008	0.443	0.184	0.518

# Panel D: Debt Granularity

			Debt	Granularity	1	
Variable	(a)	(b)	(c)	(d)	(e)	(f)
Firm FE	1.000	•	•	0.987		0.939
Year FE			1.000	0.013	0.027	0.017
Firm Size					0.040	0.008
Profitability					0.000	0.000
Tangibility					0.778	0.035
Firm Growth					0.001	0.000
Capital Expenditures					0.007	0.001
Dummy Medium Creditworthiness					0.010	0.000
Dummy High Creditworthiness					0.001	0.000
Industry Median Debt Granularity					0.014	0.000
Industry FE	•	1.000	•		0.124	
Adj. R²	0.566	0.043	0.002	0.575	0.285	0.643

#### Table 6: The influence of founder-CEO departure

The sample consists of 4,962 Belgian firms founded between 1996 and 1998 that are followed for up to 15 years after startup. Coefficients in all specifications are estimated using Firm FE. To facilitate the comparison of the coefficients, we standardize the independent and the control variables to have zero mean and unit variance. Standard errors are reported in parentheses and are computed robust to both clustering at the firm level and heteroskedasticity. Variable definitions are provided in the appendix. \*\*\*, \*\*, and \* denote statistical significance at the 1%; 5% and 10% level, respectively.

		Debt	Debt	Debt
Dependent Variable: Debt Policy	Leverage	Specialization	Maturity	Granularity
	(a)	(b)	(c)	(d)
Firm Size	0.091***	0.011***	0.058***	0.090***
	(0.003)	(0.003)	(0.003)	(0.004)
Profitability	-0.019***	-0.017***	-0.005***	-0.002
	(0.001)	(0.002)	(0.001)	(0.002)
Tangibility	0.039***	-0.002	0.039***	0.100***
	(0.002)	(0.002)	(0.002)	(0.003)
Firm Growth	0.037***	0.005***	0.017***	0.015***
	(0.001)	(0.001)	(0.001)	(0.001)
Capital Expenditures	0.005***	0.001	0.006***	0.009***
	(0.001)	(0.001)	(0.001)	(0.001)
Dummy Medium Creditworthiness	0.001	0.003**	-0.002**	0.006***
	(0.001)	(0.001)	(0.001)	(0.002)
Dummy High Creditworthiness	-0.025***	0.004**	-0.005***	-0.002
	(0.001)	(0.002)	(0.002)	(0.002)
Industry Median Debt Policy	-0.001	0.005***	0.003**	0.010***
	(0.001)	(0.002)	(0.001)	(0.003)
Founder-CEO Departure	0.097***	0.036***	0.019***	-0.040***
	(0.005)	(0.005)	(0.001)	(0.003)
Initial Debt Policy * Founder-CEO Departure	-0.093***	-0.039***	-0.033***	-0.057***
	(0.005)	(0.003)	(0.002)	(0.003)
Year FE	Yes	Yes	Yes	Yes
Firm FE	Yes	Yes	Yes	Yes
Adjusted R <sup>2</sup>	0.720	0.449	0.508	0.652
Observations	44,456	43,332	43,332	43,332

#### Table 7: The influence of founder-CEO death

The sample consists of 4,962 Belgian firms founded between 1996 and 1998. For 19 firms the founder-CEO dies as the firms are tracked for up to 15 years after startup. Coefficients in all specifications are estimated using Firm FE. To facilitate the comparison of the coefficients, we standardize the independent and the control variables to have zero mean and unit variance. Standard errors are reported in parentheses and are computed robust to both clustering at the firm level and heteroskedasticity. Variable definitions are provided in the appendix. \*\*\*, \*\*, and \* denote statistical significance at the 1%; 5% and 10% level, respectively.

		Debt	Debt	Debt
Dependent Variable: Debt Policy	Leverage	Specialization	Maturity	Granularity
	(a)	(b)	(c)	(d)
Firm Size	0.093***	0.012**	0.062***	0.092***
	(0.061)	(0.005)	(0.004)	(0.006)
Profitability	-0.020***	-0.018***	-0.006***	-0.003
	(0.002)	(0.002)	(0.001)	(0.002)
Tangibility	0.038***	-0.001	0.039***	0.101***
	(0.003)	(0.003)	(0.003)	(0.004)
Firm Growth	0.038***	0.005***	0.017***	0.015***
	(0.001)	(0.001)	(0.001)	(0.002)
Capital Expenditures	0.005***	0.001	0.005***	0.009***
	(0.001)	(0.001)	(0.001)	(0.001)
Dummy Medium Creditworthiness	0.002	0.003*	-0.002	0.006***
	(0.001)	(0.002)	(0.001)	(0.002)
Dummy High Creditworthiness	-0.025***	0.003	-0.005**	-0.001
	(0.002)	(0.002)	(0.002)	(0.003)
Industry Median Debt Policy	-0.001	0.005**	0.004**	0.008**
	(0.001)	(0.002)	(0.002)	(0.021)
Founder-CEO Death	0.031***	0.003	0.002**	-0.012***
	(0.007)	(0.004)	(0.001)	(0.004)
Initial Debt Policy * Founder-CEO Death	-0.029***	-0.002	-0.003***	-0.015***
	(0.007)	(0.005)	(0.001)	(0.004)
Year FE	Yes	Yes	Yes	Yes
Firm FE	Yes	Yes	Yes	Yes
Adjusted R <sup>2</sup>	0.716	0.446	0.500	0.649
Observations	44,456	43,332	43,332	43,332